Author’s response to reviews

**Title:** Androsterone Glucuronide to dehydroepiandrosterone sulphate ratio is discriminatory for obese Caucasian women with Polycystic Ovary Syndrome

**Authors:**

Li Wei Cho (liweicho@hotmail.com)

Thozhukat Sathyapalan (Thozhukat.Sathyapalan@hyms.ac.uk)

Eric Kilpatrick (ekilpatrick@sidra.org)

Brian Keevil (Brian.Keevil@UHSM.NHS.UK)

Adrian Miller (adrian.miller@UHSM.NHS.UK)

Anne Marie Coady (anne.coady@hey.nhs.uk)

Lina Ahmed (lha2002@qatar-med.cornell.edu)

stephen atkin (sla2002@qatar-med.cornell.edu)

**Version:** 3 **Date:** 12 Apr 2017

**Author’s response to reviews:**

1. We would like to ask for clarification whether all controls were also recruited in Hull. If the controls were recruited in another county or country we may require ethics approval from a local committee in that area.

   thank you, there is a statement page 6 line 9 of the Materials and Methods that states "All subjects were recruited in Hull."

2. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

   This has been done